Author's response to reviews

Title: In Vitro Analysis of the Biocompatibility and the Antimicrobial Effect of Four Endodontic Sealers

Authors:

Ines Willershausen (willersh@uni-mainz.de)
Angelika Callaway (callaway@uni-mainz.de)
Benjamin Briseño (briseno@uni-mainz.de)
Brita Willershausen (willersh@uni-mainz.de)

Version: 2 Date: 1 July 2011

Author's response to reviews:

Manuscript: In Vitro Analysis of the Cytotoxicity and the Antimicrobial Effect of Four Endodontic Sealers

Dear Ladies and Gentlemen,
Thank you for your letter with the comments of the two Referees
Attached please find the revised manuscript, and all changes have been highlighted in red colour.
Referee 1
The choice of the test bacteria has been included in the discussion; see page 10, lines 10-14.
The conclusion has been changed according to the suggestions included in the comments, page 12; the grammatical errors have been corrected.
In the introduction, the 2nd last paragraph has been corrected and is marked.
The batch numbers of the sealers have been added.
A new figure (Figure 1) has been created, showing photographs of the analytical balance, the axiovert microscope and the luminometer.
All references have been checked for mistakes and corrected accordingly.
Referee 2
We have replaced the word “biocompatibility” with “cytotoxicity” in the title.
The conclusions have been modified.
The English language mistakes have been corrected.

Yours sincerely,

Brita Willershausen
willersh@uni-mainz.de